ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Andrews, Ryan M. et al.

Article — Published Version

Examining the causal mediating role of brain pathology on the relationship between diabetes and cognitive impairment: the Cardiovascular Health Study

Journal of the Royal Statistical Society: Series A (Statistics in Society)

Provided in Cooperation with:

John Wiley & Sons

Suggested Citation: Andrews, Ryan M. et al. (2020) : Examining the causal mediating role of brain pathology on the relationship between diabetes and cognitive impairment: the Cardiovascular Health Study, Journal of the Royal Statistical Society: Series A (Statistics in Society), ISSN 1467-985X, Wiley, Hoboken, NJ, Vol. 183, Iss. 4, pp. 1705-1726, https://doi.org/10.1111/rssa.12570

This Version is available at: https://hdl.handle.net/10419/230005

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

J. R. Statist. Soc. A (2020) **183**, *Part* 4, *pp.* 1705–1726

Examining the causal mediating role of brain pathology on the relationship between diabetes and cognitive impairment: the Cardiovascular Health Study

Ryan M. Andrews,

Johns Hopkins University, Baltimore, USA, and Leibniz Institute for Prevention Research and Epidemiology—BIPS, Bremen, Germany

Ilya Shpitser,

Johns Hopkins University, Baltimore, USA

Oscar Lopez,

University of Pittsburgh School of Medicine, USA

William T. Longstreth,

University of Washington, Seattle, USA

Paulo H. M. Chaves, Florida International University, Miami, USA

Lewis Kuller

University of Pittsburgh, USA

and Michelle C. Carlson

Johns Hopkins University, Baltimore, and Johns Hopkins Center on Aging and Health, Baltimore, USA

[Received October 2018. Final revision March 2020]

Summary. The paper examines whether *diabetes mellitus* leads to incident mild cognitive impairment and dementia through brain hypoperfusion and white matter disease. We performed inverse odds ratio weighted causal mediation analyses to decompose the effect of diabetes on cognitive impairment into direct and indirect effects, and we found that approximately a third of the total effect of diabetes is mediated through vascular-related brain pathology. Our findings lend support for a common aetiological hypothesis regarding incident cognitive impairment, which is that diabetes increases the risk of clinical cognitive impairment in part by impacting the vasculature of the brain.

Address for correspondence: Ryan M. Andrews, Leibniz Institute for Prevention Research and Epidemiology— BIPS, Achterstrasse 30, 28359 Bremen, Germany. E-mail: andrews@leibniz-bips.de

© 2020 The Authors Journal of the Royal Statistical Society: Series A (Statistics in Society)0964–1998/20/1831705 published by John Wiley & Sons Ltd on behalf of Royal Statistical Society. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, dis-

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, o tribution and reproduction in any medium, provided the original work is properly cited.

Keywords: Alzheimer's disease; Causal inference; Causal mediation analysis; Cognitive neuroscience; Public health

1. Introduction

Diabetes mellitus is a cluster of progressive metabolic diseases that typically develop in midlife from inadequate insulin secretion or insulin action, which leads to hyperglycaemia and possibly long-term damage to multiple organs, including the blood vessels (American Diabetes Association, 2014). World wide, approximately 8–14% of the population has diabetes, with the highest prevalence seen among non-Hispanic black and Hispanic populations (Menke *et al.*, 2015; Tamayo *et al.*, 2014). Prevalence estimates rise to 20–33% among those age 65 years or older (Kirkman *et al.*, 2012). *Diabetes mellitus* (otherwise known as type 2 diabetes) is by far the most common type of diabetes among middle-aged and older adults (American Diabetes Association, 2014; Ogurtsova *et al.*, 2017).

Given that up to a third of older adults have *diabetes mellitus* and that the prevalence of the disease is expected to rise over the next few decades (Kirkman *et al.*, 2012; Menke *et al.*, 2015; Ogurtsova *et al.*, 2017), an important public health question is to understand how and why it impacts multiple systems and various health-related outcomes, including cognition. Prior studies have shown an association between *diabetes mellitus* and a number of adverse cognitive outcomes, including impaired baseline cognition, accelerated cognitive decline and higher risk of both mild cognitive impairment (MCI) and Alzheimer's disease (AD), even after accounting for common risk factors like age, obesity, level of education and comorbidities (Bangen *et al.*, 2015; Cukierman *et al.*, 2005; Gregg *et al.*, 2000; Mayeda *et al.*, 2015; Palta *et al.*, 2017; Yaffe *et al.*, 2011). In fact, some consider AD to be 'type 3' diabetes, given the strong association between the two disorders (de la Monte, 2014; Kandimalla *et al.*, 2017; Leszek *et al.*, 2017; Steen *et al.*, 2005).

One hypothesis for why diabetes increases the risk of cognitive impairment is that it leads to vascular brain injuries, such as brain hypoperfusion and white matter disease, which in turn lead to cognitive impairment. In other words, the effect of diabetes on cognitive impairment may be mediated through cerebrovascular pathology. Support for this hypothesis comes from several observational studies finding that type 2 diabetics have increased risk of stroke (Carson et al., 2012; Khoury et al., 2013; Mortel et al., 1990; Ottenbacher et al., 2004), brain hypoperfusion (Falvey et al., 2013; van Harten et al., 2006) and white matter disease (de Bresser et al., 2010; Reijmer et al., 2013). All these vascular risk factors have been shown to be strongly related to incident cognitive impairment (Gorelick et al., 2011; Kuller et al., 2004; Troncoso et al., 2008) in line with the 'vascular' hypothesis of Alzheimer's disease (de la Torre, 2012; de la Torre and Mussivand, 1993). There have also been molecular biology and animal studies that provide experimental evidence suggesting that cardiovascular disease lies along a causal pathway from diabetes mellitus to cognitive impairment. For example, Zuloaga et al. (2016) randomized mice to either a high fat diet designed to model diabetes mellitus or a low fat control diet, and then half of the mice in each group received a surgical intervention that results in brain hypoperfusion. They found that both high fat diet and brain hypoperfusion increased the risk of cognitive impairment in mice, but that the high fat diet's effect on cognitive impairment could not be explained by brain hypoperfusion alone. These controlled animal models lend further support to the potential for both a direct and an indirect causal effect between diabetes mellitus and cognitive impairment through cerebrovascular brain pathology. Several other animal model studies have found comparable results (Carvalho et al., 2013; Taguchi, 2009).

Two gaps that were highlighted by the studies above, however, are

- (a) that, to our knowledge, there has not been any human population-based study that addresses the question of whether cerebrovascular pathology is a mediator of the diabetescognitive impairment relationship and
- (b) whether these relationships are causal.

With respect to the first shortcoming, several studies have shown that cerebrovascular factors are associated with increased risk of cognitive impairment among type 2 diabetics (Bruce *et al.*, 2008; Manschot *et al.*, 2006; Umegaki *et al.*, 2008, 2015); however, these studies lack the temporal ordering that is needed to investigate whether type 2 diabetics have increased risk for cognitive impairment because *diabetes mellitus* and cognition share common causes or risk factors, or whether *diabetes mellitus* directly or indirectly leads to AD-related pathological changes like brain atrophy, neuronal death and plaques and tangles. Wang *et al.* (2017) presented results that suggest that the effect of cardiovascular burden (including diabetes) on cognitive decline is mediated by brain pathology; however, they did not consider whether the same relationships hold with respect to cognitive impairment. With regard to the second gap, it is impossible to assign *diabetes mellitus* randomly to individuals in a study for ethical reasons, which means that any study of a causal mechanism must rely on observational data. This restriction typically leads clinical and public health researchers to be circumspect with their conclusions, discussing them in terms of associations or correlations even though what they truly aim to estimate is a causal relationship (see Hernán (2018) for a discussion).

In this paper, our goal is to assess the relative magnitude of the indirect causal effect of diabetes among older adults on incident MCI or dementia through two markers of vascular brain injury: brain hypoperfusion and white matter disease. The data set that was used to address these questions was the Cardiovascular Health Study (CHS): a large 10-year prospective study of US older adults that was designed to study cardiovascular risk factors in later life through clinically adjudicated cognitive outcomes and brain and blood biomarkers (Fried *et al.*, 1991; Lopez, Kuller, Fitzpatrick, Ives, Becker and Beauchamp, 2003). To decompose the potential effect of diabetes on cognitive impairment, we used the inverse odds ratio weighting (IORW) approach of Tchetgen Tchetgen (2013), which enables the estimation of direct and indirect effects that can be interpreted as causal, provided that certain assumptions are met. Our results show that, under these assumptions, brain hypoperfusion and white matter disease jointly account for approximately a third of the total effect of diabetes on cognitive impairment.

The paper is organized as follows. Section 2 describes the CHS and our sample selection procedure. Section 3 details our estimation strategy, including the details of our causal mediation analysis and the assumptions on which it relies. Section 4 presents our results. Finally, Section 5 concludes the paper with a discussion of our findings.

2. Data and sample selection

2.1. The Cardiovascular Health Study

The data for our analysis come from the CHS, which is a population-based sample of 5888 adults age 65 years or older who reside in one of four regions in the USA:

- (a) Forsyth County, North Carolina,
- (b) Sacramento County, California,
- (c) Washington County, Maryland, or
- (d) Pittsburgh, Pennsylvania.

The recruitment strategy, study design and eligibility criteria of the CHS have been described elsewhere (see Tell et al. (1993) and Fried et al. (1991)). In short, the CHS initially recruited adults

in 1989–1990 from a random sample of all Medicare eligible individuals living in census tracts near the four recruitment regions, and a racial minority supplement of 687 individuals was added in 1992–1993. All participants were asked to complete annual in-person interviews and clinical assessments from baseline until 1998–1999. In addition to the data that were provided by the participants, the CHS also gathered information on all participant hospitalizations, including international classification of diseases, ninth revision, codes and discharge summaries. All CHS participants provided written informed consent, and the institutional review boards at each participating location approved the CHS study procedures.

We restricted our analytic sample to those in the CHS Cognition Study, which was an ancillary study to the main CHS designed in 1999 to investigate risk factors for dementia and MCI. The details of the cognition study have been published elsewhere (Lopez, Kuller, Fitzpatrick, Ives, Becker and Beauchamp, 2003). In brief, all CHS participants were eligible for the cognition study if they completed a magnetic resonance imaging (MRI) examination between 1991 and 1994 and a modified mini mental status examination evaluation, which is known as '3MS'. This resulted in a possible sample of 3602 participants (Lopez, Kuller, Fitzpatrick, Ives, Becker and Beauchamp, 2003).

2.2. Diabetes mellitus

At the baseline visit, fasting blood was drawn, kept frozen at -70 °C and shipped to a central blood analysis laboratory where serum analyses were carried out on a Kodak Ektachem 700 Analyzer (Fried *et al.*, 1991; Irie *et al.*, 2008). Fasting glucose and insulin levels were assayed, using solid phase radioimmunoassay and serum-based standards (Irie *et al.*, 2008). Baseline diabetes was defined as a fasting glucose concentration of 126 mg dl⁻¹ or higher, or the reported use of hypoglycaemic medication or insulin treatment. Participants who were determined to be prediabetic were considered non-diabetic for this analysis.

2.3. Clinical cognitive impairment

For our study, the primary outcome of clinical cognitive impairment was a diagnosis of either MCI or dementia, which were adjudicated by a panel of clinical and content experts (Lopez, Jagust, DeKosky, Becker, Fitzpatrick, Dulberg, Breitner, Lyketsos, Jones, Kawas, Carlson and Kuller, 2003; Lopez et al., 2005; Lopez, Kuller, Fitzpatrick, Ives, Becker and Beauchamp, 2003). In 1999 at the end of the clinic visits for CHS, individuals were retrospectively adjudicated for dementia starting from the date of their first MRI scan through to the end of the study. Diagnoses were made using the participants' cognitive examination, assessment of functional status, proxy interviews, physician questionnaires, telephone interviews and medical chart review, and all participants (including those who died during the study) were adjudicated if they met the inclusion criteria for the cognition study. Dementia diagnostic criteria were based on impairments in two or more cognitive domains and a history of normal intellectual function, whereas MCI diagnostic criteria were based on impairments in at least one cognitive domain, subjective cognitive complaints (either by the participant or by the participant's family members) and the absence of neurological, psychiatric or systemic illnesses that could explain the presence of cognitive deficits. Following other CHS dementia studies (Irie et al., 2008; Lyketsos et al., 2002; Podewils et al., 2005), MCI and dementia outcomes were merged into a single diagnostic entity.

2.4. Brain hypoperfusion and white matter disease

We considered two markers of vascular brain injury as mediators: brain hypoperfusion and white matter disease. Brain hypoperfusion was defined as one or more of incident stroke,

incident transient ischaemic attack or MRI findings of infarcts or extensive white matter signal abnormality. When a stroke or transient ischaemic attack was reported during followup, a provisional diagnosis was assigned by the relevant CHS field investigator, reviewed by the CHS co-ordinating centre and finally reviewed and adjudicated at annual meetings of the CHS Morbidity Review Committee (Longstreth *et al.*, 2001). Brain infarcts were identified on brain MRI and were defined on the basis of MRI signal characteristics and size 3 mm or more (Bernick *et al.*, 2001; Bryan *et al.*, 1994).

The presence of white matter disease was defined by using a semiquantitative 10-point white matter grading scale developed in the CHS (Manolio *et al.*, 1994; Yue *et al.*, 1997) designed to estimate the total extent of periventricular and subcortical white matter signal abnormality. A score of 0 or 1 represented no or barely detectable change in white matter signal, whereas a score of 9 indicated that nearly all the white matter was involved. Following prior CHS studies, we dichotomized participants as having extensive white matter disease if their white matter grade was between 3 and 9 (see Kuller *et al.* (2004, 2016) and Rosano *et al.* (2015)).

2.5. Covariates

In these analyses, we adjusted for possible confounding factors, including age, race, male sex, total cholesterol level, level of education, the presence of one or more apolipoprotein E4 (allele APOE-4), smoking status, hypertension, body mass index and antihypertensive medication use. All confounding factors were considered at their baseline values. Race was treated as a binary variable (white *versus* non-white), and APOE-4 status was also binary based on the presence of at least one allele. Smoking status was defined as current, former or never smoked. Hypertension was defined by systolic blood pressure of at least 140 mm Hg or diastolic blood pressure of at least 90 mm Hg, and antihypertensive medication use was defined as taking any antihypertensive medication. Following related prior work in the CHS (see Armstrong *et al.* (2017, 2018)), education was dichotomized as having completed high school *versus* not. All other confounding variables were treated as continuous.

2.6. Final analytic sample

2.6.1. Primary analysis

Because we sought to examine how diabetes impacts incident MCI and dementia through brain hypoperfusion, we excluded any CHS Cognition Study participants who had prevalent dementia (N = 227) or a history of stroke or transient ischaemic attack (N = 94) at baseline. In addition, we excluded participants who did not have a baseline diabetic status (N = 24). Finally, we excluded any participant who developed MCI or dementia within 3 years of baseline (N = 32), to avoid misspecifying the temporal order of events for our mediation analysis (i.e. ensuring that brain hypoperfusion and white matter disease precedes MCI or dementia). Thus, our eligible sample size was 3225 individuals. In addition, excluding individuals with missing data on one or more covariates resulted in a final analytic sample of 2872 participants. The 10% who were excluded for missing data were primarily missing APOE-4 status (268 out of 332, or 80.7%) and were statistically comparable with those who had complete data.

2.6.2. Secondary analysis

In addition to the primary analysis, we conducted two secondary analyses where we limited the outcome to either MCI or dementia rather than their combination. For the analysis involving MCI only, we excluded those who were classified with incident dementia after all other exclusion

criteria had been applied (N = 442) and, for the analysis involving dementia only, we excluded MCI cases (N = 541) after all other exclusion criteria had been applied. This resulted in smaller eligible sample sizes: 2783 for the MCI analysis and 2684 for the dementia analysis. After removing individuals with missing data, our analytic sample sizes for the secondary analyses were 2527 for MCI and 2409 for dementia.

3. Estimation strategy

3.1. Counterfactual notation and average causal effects

We are interested in estimating the causal effect of *diabetes mellitus* on cognitive impairment. To do this, we utilize the counterfactual notation of Neyman (1923) and Rubin (1974). Assume that our sample consists of N individuals, whom we can index by the subscript i = 1, ..., N, and, for each individual, we denote his diabetic status by $A_i = 1$ for diabetic and $A_i = 0$ for non-diabetic. In this paper, we refer to A_i interchangeably as either a treatment or exposure. Furthermore, let Y_i denote an individual's cognitive impairment status, with $Y_i = 1$ indicating cognitive impairment and $Y_i = 0$ indicating no impairment, and assume that we collected a vector of covariate information for each individual, denoted by C_i . We can define an individual's *counterfactual outcome* as $Y_i(A_i = a)$, which in our case defines what an individual's cognitive impairment status would be if he had (possibly counter to fact) had diabetes (i.e. $Y_i(A_i = 1)$) or if he had (possibly counter to fact) not had diabetes (i.e. $Y_i(A_i = 0)$). A causal effect of A_i on Y_i exists if $Y_i(A_i = 1) \neq Y_i(A_i = 0)$. Under the assumption that sampled individuals are independent and identically distributed, we treat A, Y and C as random variables and suppress the index i. Obviously, an individual either has or does not have *diabetes mellitus* in real life, which means that both Y(A = 1) and Y(A = 0) cannot be observed in real data. This leads many researchers to focus on population level causal effects rather than individual level effects, and we say that an average causal effect (ACE) exists in a population if $E\{Y(A=1)\} \neq E\{Y(A=0)\}$, where E is the expectation operator. More generally, the ACE can be defined as

$$ACE = h^{-1}[E\{Y(1)\}] - h^{-1}[E\{Y(0)\}]$$
(1)

where h^{-1} is a user-specified link function. If equation (1) is non-zero, this indicates the existence of an effect of the exposure on the outcome. When h^{-1} is the identity link, equation (1) is simply a linear contrast of conditional expectations; however, equation (1) also enables the ACE to be estimated by generalized linear models with non-linear link functions, like logistic regression.

To identify the ACE by using observed data, it suffices to consider four assumptions. The first is the consistency assumption, which states that, if A = a, then Y(a) = Y or, in other words, that the observed outcome Y when A = a is equal to the counterfactual outcome Y(a) (Cole and Frangakis, 2009; Pearl, 2010; VanderWeele, 2009). The second assumption is conditional exchangeability, which states that $Y(a) \perp A \mid C$ for all A. In other words, counterfactual outcomes are independent of exposure within levels of covariates C. Implicit in this assumption is also the assumption of no measurement error. The third assumption is positivity, which requires that, for a discrete exposure A and confounder vector C, if $f(C) \neq 0$ then $Pr(A = a \mid C) > 0$ for all possible values of A, where f(C) is the probability density function of C (Hernan and Robins, 2006; Westreich and Cole, 2010). In essence, this means that, within every combination of observed confounders, there are both exposed and unexposed individuals. Finally, we assume non-interference between individuals (Hudgens and Halloran, 2008; Ogburn and VanderWeele, 2014). This means that we assume that one individual's counterfactual outcomes are unaffected by another individual's treatment assignment. Under these assumptions, the ACE can be validly

estimated by using any number of approaches, including regression analysis, provided that the exposure–outcome relationship is correctly modelled.

3.2. Natural direct and indirect effects

Because we were interested in studying two possible mediators on the causal pathway between *diabetes mellitus* and cognitive impairment, we define the counterfactual value of the mediator to be $M(a) = \{M_1(a), M_2(a)\}$, with an intuitive meaning that is similar to Y(a) defined earlier and M_1 and M_2 denoting brain hypoperfusion and white matter disease respectively. For example, in our study, M(1) represents the value of M that would be observed if an individual had been (possibly counter to fact) a type 2 diabetic. When defining mediation effects, it is also common to use counterfactuals of the form $Y\{a, M(a)\}$ in which the treatment and the mediator are 'nested' together. This notation is useful, since it permits differing treatment assignments for the outcome and the mediator, e.g. $Y\{1, M(0)\}$ for the counterfactual outcome in which A = 1 for the purposes of the outcome and A = 0 for the purposes of the mediator. Using this notation, we can define two effects—the natural direct effect (NDE) and the natural indirect effect (NIE). On the mean scale, the NDE and NIE can be written as

NDE =
$$h^{-1}(E[Y\{1, M(0)\}]) - h^{-1}(E[Y\{0, M(0)\}]),$$
 (2)

NIE =
$$h^{-1}(E[Y\{1, M(1)\}]) - h^{-1}(E[Y\{1, M(0)\}]).$$
 (3)

The ability for mediation effects to be identified without regard to exact model specification is a key advantage of the causal approach to mediation, compared with other approaches, like structural equation modelling, which rely on strong parametric assumptions for identification of effects (Shpitser, 2013). Intuitively, the NDE corresponds to the effect of the exposure on the outcome if the pathway through the mediator were somehow disabled (in Fig. 1, this would mean that paths 1 and 2 would be disabled), whereas the NIE corresponds to the effect of the exposure of the exposure on the outcome if the pathway directly from the exposure to the outcome were disabled (in Fig. 1, this would mean that path 3 would be disabled). If there are no exposure-mediator interactions, the NDE and NIE sum to the ACE.

To identify these effects from observational data, it suffices to consider additional assumptions (Pearl, 2001, 2012; Robins and Greenland, 1992).

Assumption 1. $Y(a,m) \perp A|C$. Assumption 2. $M(a) \perp A|C$. Assumption 3. $Y(a,m) \perp M|A, C$. Assumption 4. $Y(a,m) \perp M(a')|C$.

Assumptions 1–3 correspond to no unmeasured confounding of the treatment–outcome relationship, no unmeasured confounding of the treatment–mediator relationship and no unmeasured confounding of the mediator–outcome relationship respectively. The fourth assumption is arguably the least intuitive and corresponds to an assumption that there is an independence across counterfactual worlds, namely that the counterfactual for the outcome in a world where exposure is set to A = a is independent of the counterfactual for the mediator in a world where exposure is set to A = a'. This 'cross-world' assumption will be usually satisfied if there is no mediator–outcome confounder that is affected by exposure (Richardson and Robins, 2013). More generally, the cross-world assumption is satisfied if we view Y(a, m, x) and M(a'x') as structural equations $f_Y(a, m, x, \epsilon_Y)$ and $f_m(a', x', \epsilon_M)$ (where X is a possible mediator–outcome



Fig. 1. Directed acyclic graph of our mediation model: the indirect effect of diabetes on MCI or dementia through brain hypoperfusion and white matter disease (paths 1 and 2) and the direct effect of diabetes on MCI or dementia (path 3) will be estimated as part of the analysis



Fig. 2. Hypothetical observed data graph in which the assumption of no mediator–outcome confounder that is affected by the exposure holds (- - \rightarrow , absence of an arrow): the letters *C* and *X* represent confounders, whereas the letter *A* denotes a treatment, *M* denotes a mediator and *Y* denotes an outcome

confounder) and have that ϵ_Y and ϵ_M are marginally independent regardless of how the inputs to the structural equations are set. This assumption is perhaps best understood by looking at a graph. In Fig. 2, we see that the variable X is a confounder of the mediator M and the outcome Y. The dotted arrow from the treatment A to X indicates that we are assuming that A does not affect X. If this arrow is present, we have a violation of this assumption because ϵ_Y and ϵ_M will not be marginally independent. Since this can never be verified or disproven experimentally, it is a strong assumption. However, we refer the reader to the discussion in Robins and Richardson (2010) for an intervention-based reconceptualization of mediation analysis under which this assumption could be tested. When satisfied, assumptions 1–4 are sufficient to allow for the estimation of NDEs and NIEs by using observed data and using convenient methods, including parametric regression. Since we are considering a joint mediator, we are assuming that there is a set of covariates C such that assumptions 1–4 are satisfied for brain hypoperfusion and white matter disease jointly.

3.3. Inverse odds ratio weighting

For our study, we estimated the NDE and NIE of *diabetes mellitus* on cognitive impairment through brain hypoperfusion and white matter disease by using IORW (Tchetgen Tchetgen, 2013). The IORW approach is a flexible framework for estimating mediation effects that can handle multiple mediators and can be used to decompose the total effect of an exposure into an NDE and NIE for most regression models that are used in policy or epidemiologic research (Nguyen *et al.*, 2015). At a high level, this is done by first estimating the total effect of the exposure on the outcome, then estimating the NDE through a weighted regression and then estimating the NIE as the difference on the additive scale between the total effect and NDE (and then transforming these estimates back to their original scale, if applicable, like with logistic or

Poisson regression). Formally, the method assumes that the total effect of A on Y, γ_{tot} , can be estimated by fitting a mean regression model

$$h^{-1}[E\{Y|A=a,C=c;\psi\}] = \tilde{\gamma}_{\text{tot}}(c;\psi_{\text{tot}})a + \tilde{\gamma}^{a'}(c;\psi_{a'})$$

where by consistency and conditional exchangeability $\tilde{\gamma}_{tot}(c; \psi_{tot}) = \gamma_{tot}(c)$ is a parametric model for the total effect with unknown parameter ψ_{tot} , $\tilde{\gamma}^{a'}(c; \psi_{a'}) = h^{-1}[E\{Y(a')|C\}]$ is a parametric model for the mean of Y(a') with unknown parameter $\psi_{a'}$ and $\psi = (\psi_{tot}^T, \psi_{a'}^T)$. It further assumes that ψ can be estimated by $\hat{\psi}$, which solves the estimating equation

$$0 = P_n[\Delta_{\text{tot}}(A, C; \hat{\psi}) \{Y - E(Y|A = a, C = c; \hat{\psi})\}]$$

where Δ_{tot} is a vector of size dim(ψ) and P_n denotes the empirical average operator. A convenient estimator for $\Delta_{tot}(a, c; \psi)$ is the maximum likelihood estimator. Furthermore, the IORW method assumes that the NDE, β_{dir} , can be estimated from

$$h^{-1}(E[Y\{a, M(a')\}|C=c; \ \beta_{\rm dir}, \psi_{a'}]) = \tilde{\gamma}_{\rm dir}(c; \ \beta_{\rm dir})a + \tilde{\gamma}^{a'}(c; \ \psi_{a'}) \tag{4}$$

where $\tilde{\gamma}_{dir}(c; \beta_{dir})$ is a parametric model for $\gamma_{dir}(c)$ with unknown parameter β_{dir} and $\beta = (\beta_{dir}, \psi_{a'})$. The key result of Tchetgen Tchetgen (2013) is that, under consistency, conditional exchangeability and positivity, plus the assumption that model (4) is correctly specified, we can obtain an unbiased estimate of β by solving the population estimating equation

$$E\{U(\beta^*)\}=0$$

where

$$U(\beta^*) = OR(M, A|C)^{-1} \Delta_{dir}(A, C; \beta^*) \{ Y - b(A, C; \beta^*) \}$$

where $b(a, c; \beta^*) = h\{\tilde{\gamma}(c; \beta^*_{dir})a + \tilde{\gamma}^{a'}(c; \psi^*_{a'})\}$ for any β^* , Δ_{dir} is defined similarly to Δ_{tot} , and

$$OR(M, A|C) = \frac{f_{M|A,C}(M|A, C) f_{M|A,C}(M = m_0|A = 0, C)}{f_{M|A,C}(M = m_0|A, C) f_{M|A,C}(M|A = 0, C)}$$
$$= \frac{f_{A|M,C}(A|M, C) f_{A|M,C}(A = 0|M = m_0, C)}{f_{A|M,C}(A = 0|M, C) f_{A|M,C}(A|M = m_0, C)}$$
(5)

where $f_{A|M,C}$ is the conditional density of A given M and C, and m_0 is a reference value for M. The equality in expression (5) comes from the fact that the odds ratio (OR) is symmetric and, since no distributional assumptions are placed on $f_{A|M,C}$, this OR parameterization is also valid for non-binary exposures. After estimating the total effect and NDE, the NIE (or joint NIE if there are multiple mediators) can be estimated by $\hat{\gamma}_{tot} - \hat{\beta}_{dir}$, and confidence intervals for all effects can be estimated by the non-parametric bootstrap or directly via variance–covariance matrix estimation. We refer readers who are interested in a more detailed description of the method to Tchetgen Tchetgen (2013).

Therefore, by using IORW, we can obtain an unbiased estimate of the NDE of an exposure on an outcome through an arbitrary number of mediators via weighted regression, with the inverse OR in equation (5) being used as the weights for those who are exposed (those who are unexposed receive a weight of 1). This is a very attractive feature of the IORW approach, since it means that it can easily handle multiple mediators through including additional terms in the OR model (one for each mediator), and continuous, binary and categorical mediators can all be included in the same model, if desired.

Applied to our research question and data, we first estimated stabilized IORW weights by fitting a logistic regression model for *diabetes mellitus* given brain hypoperfusion, white matter disease and covariates:

$$logit\{P(A=1|M_1, M_2, C)\} = \beta_0 + \beta_1 M_1 + \beta_2 M_2 + \delta' C.$$
 (6)

Although we did not consider any interactive terms in our analysis because they were considered clinically unlikely or were found to be statistically non-significant, interactions could be considered in the specification of model (6) if they are believed to be present, since the statistical consistency of the final estimators for the NDE and NIE relies on a correctly specified OR model. Using model (6), we calculated a predicted OR for diabetes for each individual in our study by taking advantage of the invariance of the OR, which enabled us statistically to consider diabetes as the dependent variable and to treat brain hypoperfusion and white matter as independent variables even though our assumed causal relationship is that diabetes causes brain hypoperfusion and white matter disease. For those who were diabetic, we calculated an IORW by taking the inverse of their predicted OR and, for those who were not diabetic, we defined their IORW to be 1. As mentioned in Nguyen et al. (2015), more efficient estimates of the NDE and NIE can be obtained by using stabilized weights, which are obtained by multiplying each individual's predicted OR by the odds of exposure when the mediators are set to their reference value, and then inverting the result. Equivalently, one could instead estimate the predicted odds of exposure by using model (6) instead of a predicted OR and then taking the inverse to obtain an inverse odds weight for each individual, and this is the stabilization method that we used.

After calculating inverse odds weights for each individual, we fitted a conditional logistic model for the total effect of diabetes on MCI or AD conditional on covariates:

$$logit\{P(Y=1|A,C)\} = \alpha_0 + \alpha_1 A + \gamma' C.$$
 (7)

The parameter α_1 in model (7) is interpreted as our estimate of the total effect. We then fitted the same model, but used the inverse odds weights:

$$logit\{P(Y=1|A,C)\} = \tilde{\alpha}_0 + \tilde{\alpha}_1 A + \tilde{\gamma}' C.$$
(8)

Crucially, no mediator terms are included in this model. In model (8), the parameter $\tilde{\alpha}_1$ is interpreted as our estimate of the NDE. Finally, we obtained an estimate of the NIE by subtracting the NDE from the total effect:

$$NIE = \alpha_1 - \tilde{\alpha}_1. \tag{9}$$

Following other studies in the literature that used the IORW approach (see Nguyen *et al.* (2015)), we *a priori* used a cut-off of p = 0.10 and 90% confidence intervals (CIs) to determine statistical significance. Given that the IORW approach is known to produce larger standard errors (Nguyen *et al.*, 2015), and because our research focus was not on the binary question of whether there is or is not a significant NIE (in favour of quantifying how large the NIE is), we felt that this decision was appropriate. To estimate 90% CIs around our estimates, we used 1000 non-parametric bootstrap replications of model (6)–(9). All analyses were implemented in Stata 15 (Stata Corp., 2017). Readers who are interested in pursuing a similar analysis to ours using the IORW approach are referred to Nguyen *et al.* (2015), where step-by-step details are given in the appendix.

4. Results

4.1. Primary and secondary outcomes

The demographic characteristics of the analytic sample are displayed in Table 1. Approximately 12.9% of our analytic sample was classified as having prevalent diabetes at baseline. Compared with non-diabetics, diabetics were significantly less likely to be white (p < 0.01), to have at least a high school education (p < 0.01) and to have at least one APOE-4 allele (p = 0.02). Diabetics also had significantly lower total cholesterol levels (p < 0.01), and they were more likely to be male compared with non-diabetics (p < 0.01). We observed no differences between diabetics and non-diabetics in terms of baseline age or smoking status. Among those who were eligible for inclusion in our study, the participants who were included in the analytic sample and those who were excluded because of missing data did not significantly differ on any of the covariates that were considered (p > 0.1).

Table 2 displays the results of our primary causal mediation analysis examining the relationship between diabetes and clinical cognitive impairment through brain hypoperfusion and white matter disease, as well as the two secondary analyses that examined MCI and dementia separately. When combining MCI and dementia as a single end point, we found that diabetes significantly increased the odds of MCI or dementia by 51% in total (OR = 1.51; 90% CI (1.22, 1.86); p < 0.01). This effect was 34% mediated by brain hypoperfusion (indirect effect $\beta = 0.14$ *versus* total effect $\beta = 0.41$). The direct effect of diabetes on cognitive impairment was significant at the p = 0.10 level (OR = 1.31; 90% CI (1.01, 1.72); p = 0.09), but the indirect effect was not (OR = 1.15; 90% CI (0.98, 1.35); p = 0.16).

Our analyses of the secondary outcomes of MCI only or dementia only followed the same patterns as the primary analysis. Specifically, we found that diabetes significantly increased the adjusted odds of incident MCI by 54% in total (OR = 1.54; 90% CI (1.21, 1.95); p < 0.01). This effect was 38% mediated by brain hypoperfusion and white matter disease (indirect effect $\beta = 0.16$ versus total effect $\beta = 0.43$). However, neither the direct nor the indirect effects were significant at the p = 0.10 level. Finally, our analysis with dementia alone as the end point suggested that diabetes significantly increased the odds of incident dementia by 56% in total (OR = 1.56; 90% CI (1.16, 2.11); p = 0.01), and this effect was 41% mediated by brain hypoperfusion and white matter disease (indirect effect $\beta = 0.18$ versus total effect $\beta = 0.45$). Neither the direct nor the indirect effects were significant on their own.

| Characteristic | Results for non-diabetics, N = 2810 | Results for diabetics, N = 415 | p-value | |
|---|--|-----------------------------------|---------|--|
| Baseline age [†] . mean (standard deviation) | 71.8 (4.8) | 72.0 (4.7) | 0.61 | |
| White, $N(\%)$ | 2463 (87.7) | 314 (75.7) | < 0.01 | |
| Male sex, $N(\%)$ | 1099 (39.1) | 203 (48.9) | < 0.01 | |
| High school educated ^{\ddagger} , N (%) | 2179 (77.5) | 289 (69.6) | < 0.01 | |
| Total cholesterol§, mean (standard deviation) | 212.6 (37.4) | 203.1 (41.8) | < 0.01 | |
| >1 APOE-4 allele, N (%) | 635 (24.6) | 71 (18.9) | 0.02 | |
| Current smoker. $N(\%)$ | 310 (11.0) | 40 (9.6) | 0.39 | |
| Former smoker, N (%) | 1171 (41.7) | 168 (40.5) | 0.71 | |
| CES-D-score§ | 4.2 (4.1) | 4.9 (4.8) | < 0.01 | |

| Table 1. | Demographic characteristics | of the CHS | Cognition | Study eligible | e sample (<i>N</i> = 3225) |
|----------|-----------------------------|------------|-----------|----------------|-----------------------------|
|----------|-----------------------------|------------|-----------|----------------|-----------------------------|

†Age was measured in years.

[‡]High school educated was defined as completing 12 years of education versus not.

§Total cholesterol was measured in units of milligrams per decilitre.

§§CES-D ranged from 0 to 29. A score of 8 or more indicates the presence of depressive symptoms.

| | Mediators: brain hypoperfusion and white matter disease | | | |
|------------------|---|--------------|---------|--|
| | OR | 90% CI | p-value | |
| MCI or dementia† | | | | |
| Total effect | 1.51 | (1.22, 1.86) | < 0.01 | |
| Direct effect | 1.31 | (1.01, 1.72) | 0.09 | |
| Indirect effect | 1.15 | (0.98, 1.35) | 0.16 | |
| MCI‡ | | | | |
| Total effect | 1.54 | (1.21, 1.95) | < 0.01 | |
| Direct effect | 1.31 | (0.98, 1.75) | 0.13 | |
| Indirect effect | 1.18 | (0.99, 1.40) | 0.12 | |
| Dementia§ | | | | |
| Total effect | 1.56 | (1.16, 2.11) | 0.01 | |
| Direct effect | 1.30 | (0.87, 1.94) | 0.28 | |
| Indirect effect | 1.20 | (0.91, 1.58) | 0.28 | |
| | | | | |

Table 2. Results of causal mediation analysis

†A total of N = 2872 CHS participants were included in this analysis, of which 842 (29.3%) were classified as having MCI or dementia. ‡A total of N = 2511 CHS participants were included in this analysis,

A total of N = 2511 CHS participants were included in this analysi of which 481 (19.2%) were classified as having MCI.

§A total of N = 2391 CHS participants were included in this analysis, of which 361 (15.1%) were classified as having dementia.

4.2. Sensitivity analyses

To assess our causal assumptions, we conducted several sensitivity analyses (Table 3). For convenience, all sensitivity analyses focused only on the primary outcome and CIs of estimates. First, we assessed how changing our definition of white matter disease would affect our results by changing the cut-off for disease by one point in either direction, and we found, not surprisingly, that raising the threshold weakened the indirect effect. Next, we changed our model specifications from logistic regression to Poisson regression while keeping the covariates in the model the same. As expected, we found that the risk ratios that were estimated by Poisson regression were closer to the null than the ORs that were estimated by logistic regression; however, the NDE and NIE estimates were qualitatively similar to each other. We also ran several additional sensitivity analyses in which we changed the definition of level of education, changed the model specification to include interaction terms between diabetes status and age, sex and race, or changed the model specification to include non-linear terms for cholesterol level, age and body mass index. In all these sensitivity analyses, our NDE and NIE estimates were fairly consistent (the results are not shown). Finally, we assessed how our NDE and NIE estimates would change if we used another mediation approach that also handles multiple mediators and estimates joint mediated effects-namely using weighted parametric regression models (VanderWeele and Vansteelandt, 2014) and natural effects models (Steen et al., 2017b) (Table 4). The method of VanderWeele and Vansteelandt (2014) produced similar results to the IORW approach; however, the natural effects modelling approach yielded a noticeably higher NDE estimate, and therefore noticeably lower NIE estimate, compared with the IORW approach.

We also ran several sensitivity analyses that were related to the key set of assumptions that we made regarding no unmeasured confounding. To assess the possibility of unmeasured confounding by omitted observed variables, we included five additional covariates in our model as

| Table 3. Model-based sensitivity analyse |
|--|
|--|

| | Mediators: brain hypoperfusion and white matter disease | | | |
|----------------------|--|--------------|------------|--------------|
| | OR | 90% CI | Risk ratio | 90% CI |
| Model 1† | | | | |
| Total effect | 1.51 | (1.22, 1.86) | 1.27 | (1.13, 1.42) |
| Direct effect | 1.31 | (1.01, 1.72) | 1.20 | (1.03, 1.39) |
| Indirect effect | 1.15 | (0.98, 1.35) | 1.06 | (0.97, 1.16) |
| Model 2 [±] | | ()) | | × , , , |
| Total effect | 1.51 | (1.22, 1.86) | 1.27 | (1.13, 1.42) |
| Direct effect | 1.34 | (1.02, 1.75) | 1.20 | (1.04, 1.40) |
| Indirect effect | 1.13 | (0.96, 1.32) | 1.05 | (0.96, 1.15) |
| Model 3§ | | | | × ´ ´ |
| Total effect | 1.51 | (1.22, 1.86) | 1.27 | (1.13, 1.42) |
| Direct effect | 1.38 | (1.05, 1.81) | 1.22 | (1.05, 1.42) |
| Indirect effect | 1.09 | (0.93, 1.29) | 1.04 | (0.95, 1.14) |
| Model 4§§ | | | | |
| Total effect | 1.54 | (1.22, 1.95) | 1.27 | (1.12, 1.45) |
| Direct effect | 1.30 | (0.97, 1.75) | 1.18 | (1.00, 1.39) |
| Indirect effect | 1.18 | (0.99, 1.42) | 1.08 | (0.98, 1.20) |

†Model inputs were kept the same as in the original analysis.

[‡]Model inputs were kept the same as in the original analysis, except that the white matter disease cut-off changed to greater than 1 out of 9 points. §Model inputs were kept the same as in the original analysis, except that the white matter disease cut-off changed to greater than 3 out of 9 points. §Model inputs were kept the same as in the original analysis, plus additional terms for baseline log₁₀(interleukin-6 level), baseline log₁₀(C-reactive protein level) depressive symptoms, income level and physical activity.

potential confounders of the relationships between diabetes, brain hypoperfusion, white matter disease and/or cognitive impairment:

- (a) a baseline $\log_{10}(\text{interleukin-6 level})$,
- (b) baseline \log_{10} (C-reactive protein level),
- (c) depressive status (measured via the Center for Epidemiological Studies depression scale CES-D),
- (d) income level and
- (e) physical activity (measured via the self-reported number of blocks walked in the month before baseline).

The inclusion of these additional covariates made the estimated NIE slightly larger, while also making both the NDE and the NIE non-significant at the p = 0.10 level (Table 3). We also ran a sensitivity analysis for unmeasured confounding by applying methods that were recommended by Tchetgen Tchetgen and Shpitser (2012). This method involves specifying a bias function for direct and indirect effects induced by unobserved confounding and iterating over a range of plausible values to see how the study results may change in the presence of various levels of unobserved confounding bias. For convenience, we ran the sensitivity analysis on participant 3MS scores rather than their adjudicated dementia status. We aimed to capture the possible bias that would result if those with diabetes who experienced brain hypoperfusion or white matter disease had, on average a potential 3MS score that was between 4 points lower and

| | Mediators: brain hypoperfusion and white matter disease | | | | | |
|---|---|------------------------------|--------------|-----------------------------|----------------------------|------------------------------|
| | IORW | | P. re | arametric gression† | Natural effects models‡ | |
| | OR | 90% CI | OR | 90% CI | OR | 90% CI |
| MCI or dementia Direct effect Indirect effect | 1.31 1.15 | (1.01, 1.72) (0.98, 1.35) | 1.40 1.11 | (1.13, 1.66) (1.0, 1.23) | 1.46 1.04 | (1.18, 1.80) (1.00, 1.07) |

Table 4. Mediation results estimated with alternative causal mediation methods

†Estimates were calculated according to the method described in VanderWeele and Vansteelandt (2014).

‡Estimates were calculated according to the method described in Steen and Vansteelandt (2019).

4 points higher compared with those who had diabetes but did not experience brain hypoperfusion or white matter disease. These end points were chosen to capture meaningful and plausible values for differences in CHS participants' cognitive performance. We observed that the direct effect and indirect effects were fairly robust to our specified levels of unmeasured confounding.

5. Discussion

5.1. Biological and clinical mechanisms

With respect to the primary goal of our analysis, we found that *diabetes mellitus* significantly increased the odds of clinical cognitive impairment by using both MCI and dementia outcomes, and that brain hypoperfusion and white matter disease mediated over a third of the total effect of diabetes on MCI and dementia. When MCI and dementia were considered separately, we found similar patterns to those with the joint outcome, with the indirect effect estimate being slightly stronger than the direct effect estimates in both cases. In general, our results fit within the broader literature showing that diabetics have a significantly greater risk of accelerated cognitive decline and cognitive impairment compared with non-diabetics (Cheng et al., 2012; Cukierman et al., 2005; Mayeda et al., 2015; Palta et al., 2017). Other studies have demonstrated that diabetics have higher cardiovascular burden, including brain lesions, that can possibly co-occur with cognitive impairment (Falvey et al., 2013; Kooistra et al., 2013; Luchsinger et al., 2007; Manschot et al., 2006; Singh et al., 2013). The study that comes closest to our own in terms of goals and result is Wang et al. (2017), which reported a significant mediating effect of brain pathology on the relationship between cardiovascular burden and cognitive decline; however, few details are available about how exactly the mediation analysis was conducted, and it is likely that they assumed strictly linear relationships between variables. Our results expand on these studies by directly evaluating this pathway by imposing a strict temporal ordering on events, and formally decomposing the effect of diabetes into both direct and indirect pathways through brain hypoperfusion and white matter disease

Our study advances epidemiologic approaches to the study of cognitive impairment in several ways. First, we used methods that enabled us to estimate more closely how a mechanistically established pathway in animal models may also be present in humans. Our study is also one of the

first, if not the first, study to examine the full hypothesized pathway in humans between diabetes, adjudicated cerebrovascular burden and adjudicated cognitive impairment in a national cohort of older adults followed for over a decade. By capitalizing on the advantages of causal mediation analysis and the CHS, we could study the pathway between diabetes and cognitive impairment through brain hypoperfusion and white matter disease in a precise temporal order. Finally, to our knowledge, this is the first study to use the IORW approach to causal mediation analysis to study a research question in clinical cognitive impairment.

At the same time, whereas we did see a statistically significant ACE and fairly sizable NDE and NIE estimates, we failed to see a significant NIE of diabetes on cognitive impairment through brain hypoperfusion and white matter disease at our *a priori* level of $\alpha = 0.10$ and, if we had used the conventional $\alpha = 0.05$ threshold, we would have failed to see a significant NDE as well (Table 2). When we repeated our analysis by using the alternative methods of VanderWeele and Vansteelandt (2014) and Steen et al. (2017b), we found similar results; however, the estimated NIE was smaller than originally estimated with IORW. Taken together, it is reasonable to be cautious when interpreting our findings. For example, although our proportion mediated was approximately a third in the main analysis, the bootstrap 90% confidence interval for this proportion is (-0.32, 0.98). Because this covers essentially every possible proportion, it is less clear what the true proportion mediated is. Another related limitation is that the proper application of mediation analysis requires that the exposure precede the mediator, which in turn precedes the outcome of interest. In our study, this important requirement meant that 10% of the total cognition study cohort was ineligible for inclusion because of our analytic exclusion criteria. It also meant that we could not examine the mediating role of other common cardiovascular factors, like hypertension, because doing so would result in losing 40-50% of the otherwise eligible sample who had a history of hypertension at baseline and would threaten our ability to detect significant effects even when they are truly present. Finally, prior studies suggest that midlife diabetes and cardiovascular risk factors are just as important, if not more important, than late life diabetes and cardiovascular risk factors (Debette et al., 2011; Gottesman et al., 2014; Rawlings et al., 2014; Tuligenga et al., 2014; Whitmer et al., 2007). In the CHS, participants were past midlife, and almost all information about cardiovascular risk factors was measured in late life. Although this does not invalidate any of the results of our study, caution should be exercised when interpreting our results in the context of the entire life course, as there are likely to be many complicated mechanisms at play and our study focused on a small subset of these.

5.2. Choice of the inverse odds ratio weighting method

When the goal of a study is to conduct a mediation analysis with multiple mediators, several options are available. However, the IORW approach has several key advantages over other comparable methods. For example, other multiple-mediator methods require that a joint density for the mediators be specified (Hong and Nomi, 2012; Lange *et al.*, 2014) or impose restrictions, like sequential causal ordering (Daniel *et al.*, 2015). The IORW approach avoids these potential complexities through the use of OR weights. Because the IORWs are calculated from a single model of the exposure given the mediators and covariates, it can handle multiple mediators in an easy straightforward way by simply including a term for each desired mediator. Another advantage of the IORW approach is that it does not require a model for the outcome given the exposure, mediator and covariates. Consequently, it is equally valid in the presence or absence of exposure-mediator interactions, unlike comparable weighting methods (e.g. the approaches of Vander-Weele and Vansteelandt (2014) and Steen *et al.* (2017b) that were used in our sensitivity analyses) that require that exposure-mediator interactions be correctly specified whenever present.

These advantages do come at a cost. Whereas other methods can produce path-specific effects

in certain situations (Daniel *et al.*, 2015; Lange *et al.*, 2014; Steen *et al.*, 2017a), the IORW approach cannot. Another disadvantage of the IORW approach is that it can produce larger variance estimates compared with other methods, which can make detecting smaller indirect effects more difficult in smaller samples (see Nguyen *et al.* (2015)). Finally, although the IORW approach requires fewer models than other mediation approaches, it still requires the correct specification of both the total effect and the OR weight model for the mediation effects to be unbiased (i.e. care needs to be taken to determine whether model (6) is correct, possibly through positing alternative models and checking how they compare and how robust the results are to model changes).

Given these advantages and disadvantages, we selected the IORW approach over other methods, primarily because of its flexibility and straightforwardness. Although it does require correct model specifications, this is a standard requirement for all methods. Since the IORW approach requires fewer models and the models that we posit are simpler objects than conditional densities, the chances of model misspecification are relatively lower compared with other methods. It would have been possibly useful to have the ability to estimate path-specific effects; however, we believe that this downside did not outweigh the other advantages that the IORW approach had for our research question.

5.3. Causal assumptions: feasibility and sensitivity analysis

To interpret our results as causal effects, it is important to assess how reasonable the assumptions that were described earlier are for our study. The more plausible the assumptions are to hold, the more confident we can be that the effects that are estimated are causal. With respect to the positivity assumption, we did find that there was a minor amount of non-overlap in some individual covariate distributions (e.g. age) between those with diabetes *versus* not, and it is possible that there were certain covariate combinations that also were unique to either the diabetes group or the control group. However, we do not feel that these practical positivity violations considerably altered our results.

Our consistency assumption is somewhat more controversial. A common consistency violation occurs when there are multiple versions of the treatment, which makes counterfactuals ill defined. When the treatment is a health condition, like diabetes, one could imagine that there are multiple versions of the health condition (e.g. diabetes due to not exercising could be one version, and diabetes due to eating a poor diet could be another version), which would mean that consistency is questionable. However, there have been some references on this assumption that argue that what matters is whether or not the possibly multiple versions of treatment result in the same outcome (Cole and Frangakis, 2009; Pearl, 2010; VanderWeele, 2009), and we believe that it is reasonable to assume that it is the diabetes itself that impacts one's future risk for brain hypoperfusion, white matter disease and MCI or dementia and that how one developed diabetes is not relevant (i.e. we assume treatment variance irrelevance, which is then used as part of the consistency assumption).

Regarding the no-interference assumption, we believe that this is reasonable. Violations of this assumption are more common in studies where the treatment is a vaccine or educational programme and/or where there is social interaction between participants because they are room mates, schoolmates, friends, neighbours, family, etc. (Ogburn and VanderWeele, 2014; VanderWeele *et al.*, 2014). Although some CHS participants did come from the same household, which would possibly violate this assumption, we believe that any possible interference effects would not have invalidated our results.

The last two general (i.e. not specific to mediation) causal assumptions-no measurement

error and correct model specification—are somewhat less likely to hold. In any study involving socio-economic or clinical measurements, it is almost impossible to measure covariates with 100% accuracy, which means that the no measured confounding assumption is very likely to be violated. However, the CHS devoted many resources towards collecting accurate data. For example, the CHS diagnosed individuals with diabetes on the basis of objective blood biomarkers and pharmacy records instead of self-report. However, with respect to our exposure, the CHS definition of diabetes was established in a younger population than the CHS recruited; therefore, it may be that our definition of diabetes, which is valid in young or middle-aged adults, is somewhat mismeasured in older adults. With respect to our mediators and outcome, cognitive impairment status and all cardiovascular-related events were adjudicated by a panel of experts. Although these do not completely rule out measurement error, it is less likely than if the CHS had relied more heavily on self-reported health variables (Butler *et al.*, 1987). As described in Section 4, we also ran several sensitivity analyses related to how variables were defined and how our models were specified, and we failed to see any notable changes from changing how variables were defined or changing the link function of our models.

With respect to our unmeasured confounding assumptions that are specific to mediation analysis, we believe that they are also plausible; however, it is impossible for us to know truly. Since we are considering a joint mediator, we are assuming that there is a set of covariates C such that assumptions 1–4 are satisfied for brain hypoperfusion and white matter disease jointly. To assess assumptions 1–3, we performed a sensitivity analysis in which we added five extra covariates to the model. We found that our estimated effects did change slightly, and our direct effect estimates became insignificant at the $\alpha = 0.10$ level. In our opinion, this means that these covariates may have been confounders that should have been included in our main model; however, it may also be that these variables are not confounders and adjusting for them either introduced bias (by opening up new collider paths or closing relevant causal paths) or unnecessarily increased our variances, thereby making the estimates non-significant. It is also plausible that there are truly unmeasured confounders of the exposure–mediator, mediator– outcome or exposure–outcome relationship(s) that we failed to include in our models, simply because the CHS did not collect them. In our opinion, however, we did manage to adjust for most, if not all, of the relevant confounders.

Finally, the cross-world assumption 4 is impossible to test empirically because it would involve setting A to two different values simultaneously. However, we cautiously believe that this assumption is plausible in our setting. The cross-world assumption will be violated if either of two scenarios happen:

- (a) there is at least one variable L that is a mediator-outcome confounder that lies on the causal pathway between A and M (VanderWeele, 2011) or
- (b) there is an unmeasured variable U that only exists as a mediator–outcome confounder under cross-world treatment assignments (Robins and Richardson, 2010).

With regard to the latter scenario, it is difficult to imagine such a confounder existing for cognitive impairment and brain hypoperfusion and white matter disease, so we believe that it does not apply. With regard to the possibility that there is at least one mediator–outcome confounder affected by exposure, we cannot be certain that this does not apply. For example, it could be that diet is such a confounder, since diabetics could change their diet after their diagnosis, and diet has been shown to be associated with both cerebrovascular pathology and cognitive impairment (Esposito *et al.*, 2010; Psaltopoulou *et al.*, 2013). To assess how a violation of the cross-world assumption would affect our results, we used the sensitivity analysis technique of Tchetgen and Shpitser (2012) which involves specifying various levels of unmeasured confounding bias (from,

for example, omitting an unmeasured mediator–outcome confounder) and iterating over them to see how the estimates of the NDE and NIE would change as a function of this bias (see the appendix of Nguyen *et al.* (2015) for an accessible step-by-step description of this sensitivity analysis). We found that under realistic amounts of bias our results did not change. This supports the belief that our results are robust against a possible cross-world assumption, even though we cannot fully rule it out.

5.4. Future directions

Since accelerated cognitive aging and incident cognitive impairment are believed to be a complex phenomenon with possibly many mediating pathways between exposures and cognitive outcomes, this paper has shown that methods such as IORW can be useful in determining the combined relative contribution of certain hypothesized mediators. Several follow-up studies could be done motivated by our results. First, these findings merit replication in other longitudinal studies that began in midlife with similar measures and are on going, like the Baltimore Longitudinal Study of Aging (Ferrucci, 2008) and the Atherosclerosis Risk in Communities Study (ARIC Investigators, 1989). These additional data could address possible pathways between midlife risk factors and late life cardiovascular and neurocognitive health, which is not possible in the CHS. Second, it will be important to examine path-specific effects of diabetes on cognitive impairment. In our study, we estimated the mediated effect of both the brain hypoperfusion and white matter disease pathways together. However, there are methods that compute path-specific effects (Avin et al., 2005; Miles et al., 2017; Steen and Vansteelandt, 2019) that could reveal which pathway may be stronger (or not) when it comes to diabetes and cognitive impairment. Specifying the contributions of the vascular and insulin pathways to cognitive impairment will be increasingly valuable as the field strives to identify the combination of modifiable factors to prevent or delay cognitive impairment best and to improve quality of life as the population shifts towards being older.

Acknowledgements

This research was supported by contracts HHSN268201200036C, HHSN268200800007C, HHSN268201800001C, N01HC55222, N01HC85079, N01HC85080, N01HC85081, N01HC 85082, N01HC85083 and N01HC85086, and grants U01HL080295 and U01HL130114 from the National Heart, Lung, and Blood Institute, with an additional contribution from the National Institute of Neurological Disorders and Stroke. Additional support was provided by grant R01AG023629 from the National Institute on Aging. A full list of principal CHS investigators and institutions can be found at CHS-NHLBI.org.

Additional research support was provided by grants T32AG027668 (to RMA), R01AI127271-01A1 (to IS), Office for Naval Research grant N00014-18-1-2760 (to IS) and National Science Foundation grants 19-571 and 1939675 (to IS).

The content of this work is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

References

American Diabetes Association (2014) Diagnosis and classification of diabetes mellitus. *Diab. Care*, **37**, suppl. 1, S81–S90.

ARIC Investigators (1989) The Atherosclerosis Risk in Communities (ARIC) Study: design and objectives. *Am. J. Epidem.*, **129**, 687–702.

- Armstrong, N. M., Carlson, M. C., Schrack, J., Xue, Q. L., Carnethon, M. R., Rosano, C., Chaves, P. H. M. and Gross, A. L. (2018) Late-life depressive symptoms as partial mediators in the associations between subclinical cardiovascular disease with onset of mild cognitive impairment and dementia. *Am. J. Geriatr. Psychiatr.*, 26, 559–568.
- Armstrong, N. M., Carlson, M. C., Xue, Q. L., Schrack, J., Carnethon, M. R., Chaves, P. H. M. and Gross, A. L. (2017) Role of late-life depression in the association of subclinical cardiovascular disease with all-cause mortality: Cardiovascular Health Study. J. Agng Hlth, 31, 652–666.
- Avin, C., Shpitser, I. and Pearl, J. (2005) Identifiability of path-specific effects. In Proc. 19th Int. Conf. Artificial Intelligence, pp. 357–363. San Francisco: Morgan Kaufmann.
- Bangen, K. J., Gu, Y., Gross, A. L., Schneider, B. C., Skinner, J. C., Benitez, A., Sachs, B. C., Shih, R., Sisco, S., Schupf, N., Mayeux, R., Manly, J. J. and Luchsinger, J. A. (2015) Relationship between type 2 diabetes mellitus and cognitive change in a multiethnic elderly cohort. J. Am. Geriatr. Soc., 63, 1075–1083.
- Bernick, C., Kuller, L., Dulberg, C., Longstreth, W. T., Manolio, T., Beauchamp, N., Price, T. and Cardiovascular Health Study Collaboration Research Group (2001) Silent MRI infarcts and the risk of future stroke: The Cardiovascular Health Study. *Neurology*, 57, 1222–1229.
- de Bresser, J., Tiehuis, A. M., van den Berg, E., Reijmer, Y. D., Jongen, C., Kappelle, L. J., Mali, W. P., Viergever, M. A., Biessels, G. J. and Utrecht Diabetic Encephalopathy Study Group (2010) Progression of cerebral atrophy and white matter hyperintensities in patients with type 2 diabetes. *Diab. Care*, 33, 1309–1314.
- Bruce, D. G., Davis, W. A., Casey, G. P., Starkstein, S. E., Clarnette, R. M., Almeida, O. P. and Davis, T. M. (2008) Predictors of cognitive decline in older individuals with diabetes. *Diab. Care*, **31**, 2103–2107.
- Bryan, R. N., Manolio, T. A., Schertz, L. D., Jungreis, C., Poirier, V. C., Elster, A. D. and Kronmal, R. A. (1994) A method for using MR to evaluate the effects of cardiovascular disease on the brain: The Cardiovascular Health Study. Am. J. Neurradiol., 15, 1625–1633.
- Butler, J. S., Burkhauser, R. V., Mitchell, J. M. and Pincus, T. P. (1987) Measurement error in self-reported health variables. *Rev. Econ. Statist.*, **69**, 644–650.
- Carson, A. P., Muntner, P., Kissela, B. M., Kleindorfer, D. O., Howard, V. J., Meschia, J. F., Williams, L. S., Prineas, R. J., Howard, G. and Safford, M. M. (2012) Association of prediabetes and diabetes with stroke symptoms: the Reasons for Geographic and Racial Differences in Stroke (REGARDS) study. *Diab. Care*, 35, 1845–1852.
- Carvalho, C., Machado, N., Mota, P. C., Correia, S. C., Cardoso, S., Santos, R. X., Santos, M. S., Oliveira, C. R. and Moreira, P. I. (2013) Type 2 diabetic and Alzheimer's disease mice present similar behavioral, cognitive, and vascular anomalies. J. Alzmrs Dis., 35, 623–635.
- Cheng, G., Huang, C., Deng, H. and Wang, H. (2012) Diabetes as a risk factor for dementia and mild cognitive impairment: a meta-analysis of longitudinal studies. *Intern. Med. J.*, **42**, 484–491.
- Cole, S. R. and Frangakis, C. E. (2009) The consistency statement in causal inference: a definition or an assumption? *Epidemiology*, 20, 3–5.
- Cukierman, T., Gerstein, H. C. and Williamson, J. D. (2005) Cognitive decline and dementia in diabetes systematic overview of prospective observational studies. *Diabetologia*, **48**, 2460–2469.
- Daniel, R. M., De Stavola, B. L., Cousens, S. N. and Vansteelandt, S. (2015) Causal mediation analysis with multiple mediators. *Biometrics*, 71, 1–14.
- Debette, S., Seshadri, S., Beiser, A., Au, R., Himali, J. J., Palumbo, C., Wolf, P. A. and DeCarli, C. (2011) Midlife vascular risk factor exposure accelerates structural brain aging and cognitive decline. *Neurology*, 77, 461–468.
- Esposito, K., Maiorino, M. I., Ceriello, A. and Giugliano, D. (2010) Prevention and control of type 2 diabetes by Mediterranean diet: a systematic review. *Diab. Res. Clin. Pract.*, 89, 97–102.
- Falvey, C. M., Rosano, C., Simonsick, E. M., Harris, T., Strotmeyer, E. S., Satterfield, S., Yaffe, K. and Health ABC Study (2013) Macro- and microstructural magnetic resonance imaging indices associated with diabetes among community-dwelling older adults. *Diab. Care*, 36, 677–682.
- Fried, L. P., Borhani, N. O., Enright, P., Furberg, C. D., Gardin, J. M., Kronmal, R. A., Kuller, L. H., Manolio, T. A., Mittelmark, M. B. and Newman, A. (1991) The Cardiovascular Health Study: design and rationale. *Ann. Epidem.*, 1, 263–276.
- Gorelick, P. B., Scuteri, A., Black, S. E., Decarli, C., Greenberg, S. M., Iadecola, C., Launer, L. J., Laurent, S., Lopez, O. L., Nyenhuis, D., Petersen, R. C., Schneider, J. A., Tzourio, C., Arnett, D. K., Bennett, D. A., Chui, H. C., Higashida, R. T., Lindquist, R., Nilsson, P. M., Roman, G. C., Sellke, F. W. and Seshadri, S. (2011) Vascular contributions to cognitive impairment and dementia: a statement for healthcare professionals from the American Heart Association/American Stroke Association. *Stroke*, 42, 2672–2713.
- Gottesman, R. F., Schneider, A. L., Albert, M., Alonso, A., Bandeen-Roche, K., Coker, L., Coresh, J., Knopman, D., Power, M. C., Rawlings, A., Sharrett, A. R., Wruck, L. M. and Mosley, T. H. (2014) Midlife hypertension and 20-year cognitive change: the Atherosclerosis Risk in Communities Neurocognitive Study. J. Am. Med. Ass. Neurol., 71, 1218–1227.
- Gregg, E. W., Yaffe, K., Cauley, J. A., Rolka, D. B., Blackwell, T. L., Narayan, K. M. and Cummings, S. R. (2000) Is diabetes associated with cognitive impairment and cognitive decline among older women?: Study of osteoporotic fractures research group. *Arch. Intern. Med.*, **160**, 174–180.
- van Harten, B., de Leeuw, F. E., Weinstein, H. C., Scheltens, P. and Biessels, G. J. (2006) Brain imaging in patients with diabetes: a systematic review. *Diab. Care*, **29**, 2539–2548.

- Hernán, M. A. (2018) The c-word: scientific euphemisms do not improve causal inference from observational data. *Am. J. Publ. Hlth*, **108**, 616–619.
- Hernan, M. A. and Robins, J. M. (2006) Estimating causal effects from epidemiological data. J. Epidem. Commty Hlth, 60, 578–586.
- Hong, G. L. and Nomi, T. (2012) Weighting methods for assessing policy effects mediated by peer change. J. Res. Educ. Effections, 5, 261–289.
- Hudgens, M. G. and Halloran, M. E. (2008) Toward causal inference with interference. J. Am. Statist. Ass., 103, 832–842.
- Irie, F., Fitzpatrick, A. L., Lopez, O. L., Kuller, L. H., Peila, R., Newman, A. B. and Launer, L. J. (2008) Enhanced risk for Alzheimer's disease in persons with type 2 diabetes and APOE epsilon4: the Cardiovascular Health Study Cognition Study. Arch. Neurol., 65, 89–93.
- Kandimalla, R., Thirumala, V. and Reddy, P. H. (2017) Is Alzheimer's disease a type 3 diabetes?: A critical appraisal. *Biochim. Biophys. Acta Molec. Basis Dis.*, **1863**, 1078–1089.
- Khoury, J. C., Kleindorfer, D., Alwell, K., Moomaw, C. J., Woo, D., Adeoye, O., Flaherty, M. L., Khatri, P., Ferioli, S., Broderick, J. P. and Kissela, B. M. (2013) Diabetes mellitus: a risk factor for ischemic stroke in a large biracial population. *Stroke*, 44, 1500–1504.
- Kirkman, M. S., Briscoe, V. J., Clark, N., Florez, H., Haas, L. B., Halter, J. B., Huang, E. S., Korytkowski, M. T., Munshi, M. N., Odegard, P. S., Pratley, R. E. and Swift, C. S. (2012) Diabetes in older adults. *Diab. Care*, 35, 2650–2664.
- Kooistra, M., Geerlings, M. I., Mali, W. P., Vincken, K. L., van der Graaf, Y. and Biessels, G. J. on behalf of the SMART-MR Study Group (2013) Diabetes mellitus and progression of vascular brain lesions and brain atrophy in patients with symptomatic atherosclerotic disease: The SMART-MR study. J. Neurol. Sci., 332, 69–74.
- Kuller, L. H., Longstreth, W. T., Arnold, A. M., Bernick, C., Bryan, R. N. and Beauchamp, N. J. on behalf of the Cardiovascular Health Study Collaborative Research Group (2004) White matter hyperintensity on cranial magnetic resonance imaging: a predictor of stroke. *Stroke*, 35, 1821–1825.
- Kuller, L. H., Lopez, O. L., Becker, J. T., Chang, Y. and Newman, A. B. (2016) Risk of dementia and death in the long-term follow-up of the Pittsburgh Cardiovascular Health Study Cognition Study. *Alzmrs Dement.*, 12, 170–183.
- Lange, T., Rasmussen, M. and Thygesen, L. C. (2014) Assessing natural direct and indirect effects through multiple pathways. Am. J. Epidem., 179, 513–518.
- Leszek, J., Trypka, E., Tarasov, V. V., Ashraf, G. M. and Aliev, G. (2017) Type 3 diabetes mellitus: a novel implication of Alzheimer's disease. *Curr. Top. Med. Chem.*, **17**, 1331–1335.
- Longstreth, Jr, W. T., Bernick, C., Fitzpatrick, A., Cushman, M., Knepper, L., Lima, J. and Furberg, C. D. (2001) Frequency and predictors of stroke death in 5,888 participants in the Cardiovascular Health Study. *Neurology*, 56, 368–375.
- Lopez, O. L., Jagust, W. J., DeKosky, S. T., Becker, J. T., Fitzpatrick, A., Dulberg, C., Breitner, J., Lyketsos, C., Jones, B., Kawas, C., Carlson, M. and Kuller, L. H. (2003) Prevalence and classification of mild cognitive impairment in the Cardiovascular Health Study Cognition Study: Part 1. Arch. Neurol., 60, 1385– 1389.
- Lopez, O. L., Kuller, L. H., Becker, J. T., Jagust, W. J., DeKosky, S. T., Fitzpatrick, A., Breitner, J., Lyketsos, C., Kawas, C. and Carlson, M. (2005) Classification of vascular dementia in the Cardiovascular Health Study Cognition Study. *Neurology*, 64, 1539–1547.
- Lopez, O. L., Kuller, L. H., Fitzpatrick, A., Ives, D., Becker, J. T. and Beauchamp, N. (2003) Evaluation of dementia in the Cardiovascular Health Cognition Study. *Neuroepidemiology*, 22, 1–12.
- Luchsinger, J. A., Reitz, C., Patel, B., Tang, M. X., Manly, J. J. and Mayeux, R. (2007) Relation of diabetes to mild cognitive impairment. Arch. Neurol., 64, 570–575.
- Lyketsos, C. G., Lopez, O., Jones, B., Fitzpatrick, A. L., Breitner, J. and DeKosky, S. (2002) Prevalence of neuropsychiatric symptoms in dementia and mild cognitive impairment: Results from the Cardiovascular Health Study. J. Am. Med. Ass., 288, 1475–1483.
- Manolio, T. A., Kronmal, R. A., Burke, G. L., Poirier, V., O'Leary, D. H., Gardin, J. M., Fried, L. P., Steinberg, E. P. and Bryan, R. N. (1994) Magnetic resonance abnormalities and cardiovascular disease in older adults. The Cardiovascular Health Study. Stroke, 25, 318–327.
- Manschot, S. M., Brands, A. M., van der Grond, J., Kessels, R. P., Algra, A., Kappelle, L. J., Biessels, G. J. and the Utrecht Diabetic Encephalopathy Group (2006) Brain magnetic resonance imaging correlates of impaired cognition in patients with type 2 diabetes. *Diabetes*, 55, 1106–1113.
- Mayeda, E. R., Whitmer, R. A. and Yaffe, K. (2015) Diabetes and cognition. Clin. Geriatr. Med., 31, 101–115.
- Menke, A., Casagrande, S., Geiss, L. and Cowie, C. C. (2015) Prevalence of and trends in diabetes among adults in the United States, 1988–2012. J. Am. Med. Ass., **314**, 1021–1029.
- Miles, C. H., Shpitser, I., Kanki, P., Meloni, S. and Tchetgen, E. J. T. (2017) Quantifying an adherence path-specific effect of antiretroviral therapy in the Nigeria PEPFAR program. J. Am. Statist. Ass., **112**, 1443–1452.
- de la Monte, S. M. (2014) Type 3 diabetes is sporadic Alzheimer's disease: mini-review. *Eur. Neurpsychpharmcol.*, **24**, 1954–1960.
- Mortel, K. F., Meyer, J. S., Sims, P. A. and McClintic, K. (1990) Diabetes mellitus as a risk factor for stroke. Sth Med. J, 83, 904–911.

- Neyman, J. (1923) Sur les applications de la thar des probabilités aux experiences agaricales: essay des principes (Engl. transl. D. Dabrowska and T. P. Speed, *Statist. Sci.*, **5**, (1990), 463–472).
- Nguyen, Q. C., Osypuk, T. L., Schmidt, N. M., Glymour, M. M. and Tchetgen Tchetgen, E. J. (2015) Practical guidance for conducting mediation analysis with multiple mediators using inverse odds ratio weighting. Am. J. Epidem., 181, 349–356.
- Ogburn, E. L. and VanderWeele, T. J. (2014) Causal diagrams for interference. Statist. Sci., 29, 559-578.
- Ogurtsova, K., da Rocha Fernandes, J. D., Huang, Y., Linnenkamp, U., Guariguata, L., Cho, N. H., Cavan, D., Shaw, J. E. and Makaroff, L. E. (2017) IDF Diabetes Atlas: global estimates for the prevalence of diabetes for 2015 and 2040. *Diab. Res. Clin. Pract.*, **128**, 40–50.
- Ottenbacher, K. J., Ostir, G. V., Peek, M. K. and Markides, K. S. (2004) Diabetes mellitus as a risk factor for stroke incidence and mortality in Mexican American older adults. J. Gerontol. A, 59, M640–M645.
- Palta, P., Carlson, M. C., Crum, R. M., Colantuoni, E., Sharrett, A. R., Yasar, S., Nahin, R. L., DeKosky, S. T., Snitz, B., Lopez, O., Williamson, J. D., Furberg, C. D., Rapp, S. R. and Golden, S. H. (2017) Diabetes and cognitive decline in older adults: the Ginkgo Evaluation of Memory Study. J. Gerontol. A, 73, 123–130.
- Pearl, J. (2001) Direct and indirect effects. In Proc. 17th Conf. Uncertainty and Artificial Intelligence, pp. 411–420. San Francisco: Morgan Kaufmann.
- Pearl, J. (2010) On the consistency rule in causal inference: axiom, definition, assumption, or theorem? *Epidemiology*, 21, 872–875.
- Pearl, J. (2012) The causal mediation formula—a guide to the assessment of pathways and mechanisms. *Prev. Sci.*, **13**, 426–436.
- Podewils, L. J., Guallar, E., Kuller, L. H., Fried, L. P., Lopez, O. L., Carlson, M. and Lyketsos, C. G. (2005) Physical activity, APOE genotype, and dementia risk: findings from the Cardiovascular Health Cognition Study. Am. J. Epidem., 161, 639–651.
- Psaltopoulou, T., Sergentanis, T. N., Panagiotakos, D. B., Sergentanis, I. N., Kosti, R. and Scarmeas, N. (2013) Mediterranean diet, stroke, cognitive impairment, and depression: a meta-analysis. Ann. Neurol., 74, 580– 591.
- Rawlings, A. M., Sharrett, A. R., Schneider, A. L., Coresh, J., Albert, M., Couper, D., Griswold, M., Gottesman, R. F., Wagenknecht, L. E., Windham, B. G. and Selvin, E. (2014) Diabetes in midlife and cognitive change over 20 years: a cohort study. *Ann. Intern. Med.*, 161, 785–793.
- Reijmer, Y. D., Leemans, A., Brundel, M., Kappelle, L. J., Biessels, G. J. and the Utrecht Vascular Cognitive Impairment Study Group (2013) Disruption of the cerebral white matter network is related to slowing of information processing speed in patients with type 2 diabetes. *Diabetes*, 62, 2112–2115.
- Richardson, T. S. and Robins, J. M. (2013) Single world intervention graphs: a primer. 2nd Uncertainty in Artificial Intelligence Wrkshp Causal Structure Learning, Bellevue.
- Robins, J. M. and Greenland, S. (1992) Identifiability and exchangeability for direct and indirect effects. *Epidemiology*, 3, 143–155.
- Robins, J. M. and Richardson, T. S. (2010) Alternative graphical causal models and the identification of direct effects. In *Causality and Psychopathology: Finding the Determinants of Disorders and Their Cures* (eds P. E. Shrout, K. M. Keyes and K. Ornstein), pp. 103–158. New York: Springer.
- Rosano, C., Abebe, K. Z., Aizenstein, H. J., Boudreau, R., Jennings, J. R., Venkatraman, V., Harris, T. B., Yaffe, K., Satterfield, S., Newman, A. B. and Study, H. A. (2015) Longitudinal systolic blood pressure characteristics and integrity of white matter tracts in a cohort of very old black and white adults. *Am. J. Hypertens.*, 28, 326–334.
- Rubin, D. B. (1974) Estimating causal effects of treatments in randomized and non-randomized studies. J. Educ. Psychol., 66, 688–701.
- Shpitser, I. (2013) Counterfactual graphical models for longitudinal mediation analysis with unobserved confounding. Cogn Sci., 37, 1011–1035.
- Singh, G. M., Danaei, G., Farzadfar, F., Stevens, G. A., Woodward, M., Wormser, D., Kaptoge, S., Whitlock, G., Qiao, Q., Lewington, S., Di Angelantonio, E., Vander Hoorn, S., Lawes, C. M., Ali, M. K., Mozaffarian, D. and Ezzati, M. (2013) The age-specific quantitative effects of metabolic risk factors on cardiovascular diseases and diabetes: a pooled analysis. *PLOS One*, 8, no. 7, article e65174.
- Stata Corp. (2017) Stata Statistical Software: Release 15. College Station: Stata Corp.
- Steen, E., Terry, B. M., Rivera, E. J., Cannon, J. L., Neely, T. R., Tavares, R., Xu, X. J., Wands, J. R. and de la Monte, M. (2005) Impaired insulin and insulin-like growth factor expression and signaling mechanisms in Alzheimer's disease—is this type 3 diabetes? J. Alzmrs Dis., 7, 63–80.
- Steen, J., Loeys, T., Moerkerke, B. and Vansteelandt, S. (2017a) Flexible mediation analysis with multiple mediators. Am. J. Epidem., 186, 184–193.
- Steen, J., Loeys, T., Moerkerke, B. and Vansteelandt, S. (2017b) Medflex: an R package for flexible mediation analysis using natural effect models. J. Statist. Softwr., 76, no. 11, 1–46.
- Steen, J. and Vansteelandt, S. (2019) Graphical models for mediation analysis. In Handbook of Graphical Models (eds M. Maathuis et al.), pp. 405–428. Boca Raton: CRC Press.
- Taguchi, A. (2009) Vascular factors in diabetes and Alzheimer's disease. J. Alzmrs Dis., 16, 859–864.
- Tamayo, T., Rosenbauer, J., Wild, S. H., Spijkerman, A. M., Baan, C., Forouhi, N. G., Herder, C. and Rathmann, W. (2014) Diabetes in Europe: an update. *Diab. Res. Clin. Pract.*, 103, 206–217.

- Tchetgen Tchetgen, E. J. (2013) Inverse odds ratio-weighted estimation for causal mediation analysis. Statist. Med., 32, 4567–4580.
- Tchetgen, E. J. T. and Shpitser, I. (2012) Semiparametric theory for causal mediation analysis: efficiency bounds, multiple robustness, and sensitivity analysis. *Ann. Statist.*, **40**, 1816–1845.
- Tell, G. S., Fried, L. P., Hermanson, B., Manolio, T. A., Newman, A. B. and Borhani, N. O. (1993) Recruitment of adults 65 years and older as participants in the Cardiovascular Health Study. *Ann. Epidem.*, **3**, 358–366.
- de la Torre, J. C. (2012) Cardiovascular risk factors promote brain hypoperfusion leading to cognitive decline and dementia. *Cardvasc. Psychiatr. Neurol.*, article 367516.
- de la Torre, J. C. and Mussivand, T. (1993) Can disturbed brain microcirculation cause Alzheimer's disease? *Neurol. Res.*, **15**, 146–153.
- Troncoso, J. C., Zonderman, A. B., Resnick, S. M., Crain, B., Pletnikova, O. and O'Brien, R. J. (2008) Effect of infarcts on dementia in the Baltimore Longitudinal Study of Aging. Ann. Neurol., 64, 168–176.
- Tuligenga, R. H., Dugravot, A., Tabák, A. G., Elbaz, A., Brunner, E. J., Kivimäki, M. and Singh-Manoux, A. (2014) Midlife type 2 diabetes and poor glycaemic control as risk factors for cognitive decline in early old age: a post-hoc analysis of the Whitehall II cohort study. *Lancet Diab. Endocrinol.*, 2, 228–235.
- Umegaki, H., Iimuro, S., Kaneko, T., Araki, A., Sakurai, T., Ohashi, Y., Iguchi, A. and Ito, H. (2008) Factors associated with lower mini mental state examination scores in elderly Japanese diabetes mellitus patients. *Neurbiol. Agng*, 29, 1022–1026.
- Umegaki, H., Kawamura, T., Umemura, T. and Kawano, N. (2015) Factors associated with cognitive decline in older adults with type 2 diabetes mellitus during a 6-year observation. *Geriatr. Gerontol. Int.*, 15, 302–310.
- VanderWeele, T. J. (2009) Concerning the consistency assumption in causal inference. *Epidemiology*, 20, 880–883.
 VanderWeele, T. J. (2011) Controlled direct and mediated effects: definition, identification and bounds. *Scand. J. Statist.*, 38, 551–563.
- VanderWeele, T. J., Tchetgen Tchetgen, E. J. and Halloran, M. E. (2014) Interference and sensitivity analysis. Statist. Sci., 29, 687–706.
- VanderWeele, T. J. and Vansteelandt, S. (2014) Mediation analysis with multiple mediators. *Epidem. Meth.*, **2**, 95–115.
- Wang, R., Fratiglioni, L., Kalpouzos, G., Lovden, M., Laukka, E. J., Bronge, L., Wahlund, L. O., Backman, L. and Qiu, C. X. (2017) Mixed brain lesions mediate the association between cardiovascular risk burden and cognitive decline in old age: a population-based study. *Alzmrs Dem.*, 13, 247–256.
- Westreich, D. and Cole, S. R. (2010) Positivity in practice (with discussion). Am. J. Epidem., 171, 674-681.
- Whitmer, R. A., Gunderson, E. P., Quesenberry, C. P., Zhou, J. and Yaffe, K. (2007) Body mass index in midlife and risk of Alzheimer's disease and vascular dementia. *Curr. Alzmrs Res.*, 4, no. 2, 103–109.
- Yaffe, K., Lindquist, K., Schwartz, A. V., Vitartas, C., Vittinghoff, E., Satterfield, S., Simonsick, E. M., Launer, L., Rosano, C., Cauley, J. A. and Harris, T. (2011) Advanced glycation end product level, diabetes, and accelerated cognitive aging. *Neurology*, 77, 1351–1356.
- Yue, N. C., Arnold, A. M., Longstreth, W. T., Elster, A. D., Jungreis, C. A., O'Leary, D. H., Poirier, V. C. and Bryan, R. N. (1997) Sulcal, ventricular, and white matter changes at MR imaging in the aging brain: data from the Cardiovascular Health Study. *Radiology*, 202, 33–39.
- Zuloaga, K. L., Johnson, L. A., Roese, N. E., Marzulla, T., Zhang, W., Nie, X., Alkayed, F. N., Hong, C., Grafe, M. R., Pike, M. M., Raber, J. and Alkayed, N. J. (2016) High fat diet-induced diabetes in mice exacerbates cognitive deficit due to chronic hypoperfusion. J. Cereb. Blood Flow Metab., 36, 1257–1270.